Author's response to reviews

Title: Role of C-reactive protein as a biomarker for prediction of the severity of pulmonary exacerbations in patients with cystic fibrosis.

Authors:

Rosa María Girón Moreno (rmgiron@gmail.com)
Jose Luis Justicia (jljusticia@gmail.com)
Sara Yamamoto (saiux_25@hotmail.com)
Claudia Valenzuela (claudiavale@hotmail.com)
Carolina Cisneros (carol9199@yahoo.es)
Rosa Mar Gómez-Punter (rosamar_gp@hotmail.com)
Gilda Fernandes (gmfdes@hotmail.com)
Julio Ancochea (juli119@gmail.com)

Version: 2
Date: 31 May 2014

Author's response to reviews: see over
Dear Mr./Mrs.,

We’ve submitted this manuscript to you to assess its publication in Cystic Fibrosis Journal as an Original Article.

All the coauthors contributed to the acquisition, analysis, and interpretation of data. All of them also helped draft the article and reviewed it critically for important intellectual content and approved the final version.

We consider the value of CRP as a prognostic factor in exacerbations in patients with cystic fibrosis an interesting and innovative subject, since no data about it has been published yet.

Sincerely,

Rosa Mª Girón Moreno
Director of Cystic Fibrosis Unit
Instituto de Investigación La Princesa